Author’s response to reviews

Title: National and subnational hypertension prevalence estimates for the Republic of Ireland: better subnational data are needed to produce better subnational prevalence estimates

Authors:

Steve Barron (steve.Barron@publichealth.ie)
Kevin Balanda (kevin.balanda@publichealth.ie)
John Hughes (john.hughes@qub.ac.uk)
Lorraine Fahy (lorraine.fahy@publichealth.ie)

Version: 2 Date: 23 July 2013

Author’s response to reviews:

The following ethical statement is now included in the methods section of the manuscript:

'This study did not require any ethical approval as it involved secondary analyses of publicly available demographic and population data.'